Author’s response to reviews

Title: Higher preoperative pain catastrophizing increases the risk of low patient reported satisfaction after carpal tunnel release: a prospective study

Authors:
Sebastian Breddam Mosegaard (sebmos@rm.dk)
Maiken Stilling (maiken.stilling@clin.au.dk)
Torben Bæk Hansen (torbehns@rm.dk)

Version: 3  Date: 09 Jan 2020

Author’s response to reviews:

Assistant Editor Comments:

1. Abstract
   1. Answer: We have removed the trial registration section from the abstract.

2. Figures
   2. Answer: We have made sure that we are only referring to figure 1.